Author’s response to reviews

Title: Validating a transnational fracture treatment registry using a standardized method

Authors:

Jasper Frese (jasper.frese@yahoo.de)
Annalice Gode (annalice.gode@uksh.de)
Gerhard Heinrichs (gerhard.heinrichs@uksh.de)
Armin Will (armin.will@uksh.de)
Arndt-Peter Schulz (schulz@biomechatronics.de)

Version: 3 Date: 31 Oct 2019

Author’s response to reviews:

Technical Comments:

1. Please move your keywords to the end of your Abstract.
Answer: Keywords are moved.

2. Please move your abbreviations & declarations to before your references.
Answer: Both are moved.

Editor Comments:

1. Consent to participate

   -- In your “ethical approval and consent to participate” section, please confirm whether informed consent was obtained in written or verbal format. If verbal, please confirm if an ethics committee approved the procedure for verbal consent, and why they felt the need for written consent was not necessary for this study. Please also detail how you documented the consent.

   Answer: Written consent was taken. Now mentioned in the corresponding section

2. Clean copy

   -- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Answer: The remaining comments are now deleted.